Author's response to reviews

Title: A cluster randomised feasibility trial evaluating nutritional interventions in the treatment of malnutrition in care home-dwelling adults

Authors:

Ruth E Stow Miss (ruth.stow@nottingham.ac.uk)
Caroline E Rick Dr (c.e.rick@bham.ac.uk)
Christina Smith Dr (Christina.smith@ucl.ac.uk)
Natalie J Ives Miss (n.j.ives@bham.ac.uk)
Alison B Rushton Dr (a.b.rushton@bham.ac.uk)

Version: 2

Date: 14 May 2015

Editor

Trials

13th May 2015

Dear Editor

Re: A cluster randomised feasibility trial evaluating nutritional interventions in the treatment of malnutrition in care home-dwelling adults

I am revising the manuscript to change the order of authors on the first page. It was brought to my attention by one of the co-authors and the lead supervisor of this MRes project that the order should be changed to reflect supervisory roles. An email from each author agreeing to the changes will also be provided to support this change.

Please direct all correspondence pre-publication to ruth.stow@nottingham.ac.uk. I look forward to hearing from you.

Yours sincerely

Ruth Stow

Registered Dietitian

Assistant Professor in Nutrition and Dietetics
On behalf of all authors